**Author's response to reviews**

**Title:** Disease characteristics and treatment of women with early onset breast cancer participating in the Prospective study of Outcomes in Sporadic versus Hereditary breast cancer (POSH)

**Authors:**

Diana M Eccles (de1@soton.ac.uk)
Susan M Gerty (S.M.Gerty@soton.ac.uk)
Peter D Simmonds (P.D.Simmonds@soton.ac.uk)
Sarah Ennis (S.Ennis@soton.ac.uk)
Douglas G Altman (doug.altman@cancer.org.uk)
Victoria N Hammond (vnh@soton.ac.uk)
Posh Steering Group (de1@soton.ac.uk)

**Version:** 3 **Date:** 5 April 2007

**Author's response to reviews:**

I have moved the sentence about written informed consent from the "Data Collection" section to the immediately preceding "Method of Recruitment" section of the protocol description.